Author's response to reviews

Title: Trial protocol: Rapid reduction versus abrupt quitting for smokers who want to stop soon: a randomised controlled non-inferiority trial

Authors:

Nicola Lindson (nll839@bham.ac.uk)
Paul Aveyard (p.n.aveyard@bham.ac.uk)
Jackie T Ingram (j.t.ingram@bham.ac.uk)
Jennie Inglis (j.inglis@bham.ac.uk)
Jane Beach (jane.beach@sbpct.nhs.uk)
Robert West (robert.west@ucl.ac.uk)
Susan Michie (s.michie@ucl.ac.uk)

Version: 4 Date: 13 August 2009

Author's response to reviews:

The following sentence has been added to acknowledgments pg.31:

This work was undertaken by The UK Centre for Tobacco Control Studies, a UKCRC Public Health Research: Centre of Excellence. Funding from British Heart Foundation, Cancer Research UK, Economic and Social Research Council, Medical Research Council, and the Department of Health, under the auspices of the UK Clinical Research Collaboration, is gratefully acknowledged.

No formatting changes have been made.